\begin{table}%t3
\caption{Results for the brown dwarf candidates (see Sect.~\ref{bd.sec}).
\label{bd_tab}       }
%
%%\centerline
\par
 {\begin{tabular}{rrccrrrrrr}
\hline\hline\noalign{\smallskip}
 Name &Ref&  SpT &  $\log\left(\frac{\displaystyle L_{\rm bol}}{\displaystyle L_\odot}\right)$ &  
 $M$ &  $A_V$ & source & $T_{\rm X}$ & $L_{\rm X} $~~~~~~~~ &  $\log\left(\frac{\displaystyle L_{\rm X}}{\displaystyle L_{\rm bol}}\right)$ \\
      &  &    &  & $[M_\odot]$ & [mag] & counts  & [MK] & [erg/s] & \\
%
\noalign{\smallskip}\hline\noalign{\smallskip}
%
B185815.3-370435 &1&   &  &    & $\sim$7 &   $<$3.2 &  &  $<$$2.6\times 10^{27}   $ & \\
%
B185831.1-370456 &1, 2& M8.5 & $-2.6 $ & $\sim$0.025  & $\sim$0  & 9& &   $\sim$$8.0\times 10^{26}$ & $\sim$$-4.1 $ \\
%
B185839.6-365823  &1&   & $-3.1 $  & &  $\sim$13  &  47& $>$350  & $\sim$$2.7\times 10^{28}$ & $\sim$$-2.1 $\\ 
% = 190201.92--365400.4
%
B185840.4-370433  &1&   & $-2.2 $ &   &  $\sim$12&  6  &  & $\sim$$1\times 10^{28}$& $\sim$$-3.4$\\
%
B185853.3-370328 &1&   &$-2.2 $ &   &  $\sim$12& $<$2.1 & & $<$$4.5\times 10^{27}$ & $<$$-3.7$\\
%
LS-RCrA 1 &3& M6.5-7 & $-2.6$ & $\sim$0.04--0.08 & $\sim$0.5 & $<$10 &  &   $<$$1.1\times 10^{27}$ & $<$$-3.9 $\\
%
LS-RCrA 2   &3& M6   & $-1.5$  & $\sim$0.08 &  $\sim$1.2 &   113 & 4.5 &  $3.1\times 10^{28}$ & $-3.6$ \\  % Lx from fit
% = 190149.35--370028.7
%
ISO CrA 143 &4& $\approx$M8: & $-2.0  $ & $\sim$0.025 & $\sim$2.3 & 88 &  8 & $2.9 \times 10^{28}$ & $-3.2$ \\
\hline
\end{tabular}}
\par
\smallskip
References: 1:~\citet{Neuhaeuser99}; 2:~\citet{Lopez05};
3:~\citet{Fernandez01}; 4:~\cite{Olofsson99}.
\end{table}